Author’s response to reviews

Title: Transcatheter Occlusion of Giant Congenital Coronary Cameral Fistulae: Case reports

Authors:

Deogratias Arnold Nkya (gratias1980@gmail.com)
Greenwood Sinyangwe (gsinyangwe@gmail.com)
Farirai Takawira (fari.takawira@up.ac.za)

Version: 3 Date: 31 Aug 2019

Author’s response to reviews:

To

The Editor in Chief,

Journal of Medical Case Reports

Dear Sir/Madam,

REF: SUBMISSION OF THE REVISED MANUSCRIPT ENTITLED “TRANSCATHETER OCCLUSION OF GIANT CONGENITAL CORONARY CAMERAL FISTULAE: CASE REPORTS”

We want to thank you for potentially accepting our manuscript for publication in Journal of Medical Case Reports, after carrying out some essential revisions suggested by our reviewers.

I am writing to inform you that we have carried out the revisions suggested by the reviewers as outlined below;

1. We have added the detailed systemic examination which included neurologic, abdominal and respiratory examinations for all the three cases.

2. We have added important vital signs for all the three cases.
3. Follow up for the case #3 is included in this revision.

4. We have added laboratory results which included complete blood count, C reactive protein, renal function tests and serum electrolytes

5. Laboratory results for case #3 is not included in this revision because of the we could not retrieve them from our electronic system. Failure of retrieval is caused by electronic system upgrade that resulted to achieved data loss. The patient was otherwise in good clinical condition at the time of coronary cameral fistula closure.

6. Other laboratory results were not done at all at the time of preparation for percutaneous intervention because there was no clinical indications for the tests and also to reduce the procedure costs. These included liver function tests and urinalysis.

7. The manuscript was also reviewed by Language reviewer as previously required. Certificate of language review is attached.

Regards,

Deogratias A. Nkya